\begin{table} %T5
\caption{Objects in this sample with proper motions in the range $\rm 0.1''/yr<\mu<0.5''/yr$ which have estimated distances closer that 20~pc. }
         \label{close0.2}
%\centerline
 {\small
\begin{tabular}{lcccccccl}
\hline\hline
Name&Position&Distance&$\mu$&$I$&$J$&$H$&$K_{\rm S}$&Other\\
&&pc&($''$/yr)&&&&&Name\\
\hline 
\object{SIPS1848-8214} &18 48 51.35 $-$82 14 40.4&14.26$\pm^{8.50}_{5.33}$&0.272&14.373&11.482&10.922&10.503&\\ 
\object{SIPS1731-7851} &17 31 45.16 $-$78 51 23.3&16.94$\pm^{10.09}_{6.32}$&0.362&14.173&11.613&11.007&10.675&\\ 
\object{SIPS0630-7643} &06 30 46.68 $-$76 43 15.3&8.69$\pm^{5.18}_{3.25}$&0.483&10.738&8.894&8.275&7.923&\object{SCR J0630-7643} $^1$\\ 
\object{SIPS1809-7613} &18 09 06.96 $-$76 13 23.0&15.92$\pm^{9.55}_{5.97}$&0.156&11.615&9.817&9.275&8.989&\\ 
\object{SIPS2016-7531} &20 16 10.88 $-$75 31 04.8&19.06$\pm^{11.38}_{7.12}$&0.253&12.247&10.465&9.864&9.509&\\ 
\object{SIPS0504-7401} &05 04 26.15 $-$74 01 55.3&19.03$\pm^{11.36}_{7.12}$&0.359&12.147&10.346&9.778&9.469&\\ 
\object{SIPS1826-6542} &18 26 46.79 $-$65 42 37.7&11.28$\pm^{6.72}_{4.21}$&0.311&12.913&10.569&9.960&9.547&\\ 
\object{SIPS2045-6332} &20 45 02.28 $-$63 32 05.3&15.15$\pm^{9.03}_{5.66}$&0.223&15.950&12.619&11.807&11.207&\\ 
\object{SIPS0152-6329} &01 52 55.17 $-$63 29 30.2&16.67$\pm^{9.94}_{6.23}$&0.140&11.993&10.167&9.604&9.261&\\ 
\object{SIPS2019-5816} &20 19 49.82 $-$58 16 41.1&13.66$\pm^{8.14}_{5.10}$&0.347&12.868&10.664&10.104&9.715&\\ 
\object{SIPS2255-5713} &22 55 18.70 $-$57 13 04.0&12.92$\pm^{8.72}_{5.21}$&0.341&17.851&14.083&13.189&12.579&\object{2M2255-57} $^2$\\ 
\object{SIPS0128-5545} &01 28 26.76 $-$55 45 34.6&15.24$\pm^{9.79}_{5.96}$&0.293&17.447&13.775&12.916&12.336&\object{2M0128-55} $^2$\\ 
\object{SIPS0139-3936} &01 39 21.55 $-$39 36 05.8&14.99$\pm^{8.96}_{5.61}$&0.283&10.791&9.209&8.629&8.274&\\ 
\object{SIPS1110-3731} &11 10 27.98 $-$37 31 51.8&10.72$\pm^{6.41}_{4.01}$&0.103&9.041&7.651&7.041&6.774&\object{TWA 3A} $^3$\\ 
\object{SIPS0339-3525} &03 39 34.82 $-$35 25 48.4&9.95$\pm^{5.96}_{3.73}$&0.405&13.288&10.725&10.017&9.548&\object{APMPM J0340-3526}  $^4$\\ 
\object{SIPS0604-3433} &06 04 52.13 $-$34 33 38.6&13.76$\pm^{8.26}_{5.16}$&0.330&9.027&7.742&7.183&6.866&\object{LP 949-15}\\ 
\object{SIPS0124-3355} &01 24 30.52 $-$33 55 00.6&18.85$\pm^{11.28}_{7.06}$&0.222&12.445&10.555&10.009&9.682&\object{LP 939-44}\\ 
\object{SIPS1346-3149} &13 46 46.24 $-$31 49 26.7&16.03$\pm^{9.58}_{6.00}$&0.366&13.165&10.975&10.436&10.038&\object{LP 911-56}\\ 
\object{SIPS1037-2746} &10 37 45.41 $-$27 46 40.2&18.13$\pm^{10.86}_{6.79}$&0.322&9.936&8.595&8.035&7.719&\object{CE 186}\\ 
\object{SIPS2351-2537} &23 51 50.25 $-$25 37 37.8&18.35$\pm^{10.94}_{6.86}$&0.420&15.573&12.471&11.725&11.269&\object{LEHPM 6334}\\ 
\object{SIPS1625-2508} &16 25 49.66 $-$25 08 17.8&8.62$\pm^{16.80}_{5.70}$&0.147&17.800&13.745&13.117&12.728&\\ 
\object{SIPS1042-2416} &10 42 41.32 $-$24 16 06.2&17.13$\pm^{10.21}_{6.40}$&0.205&12.087&10.277&9.672&9.338&\object{NLTT 25128}\\ 
\object{SIPS1504-2355} &15 04 16.35 $-$23 55 55.8&19.12$\pm^{11.41}_{7.15}$&0.327&14.652&12.011&11.383&11.032&\object{LP 859-1}\\ 
\object{SIPS1309-2330} &13 09 21.85 $-$23 30 33.9&15.64$\pm^{9.35}_{5.85}$&0.383&14.514&11.785&11.082&10.669&\object{CE 303}\\ 
\object{SIPS1155-2224} &11 55 42.94 $-$22 24 58.2&13.50$\pm^{8.05}_{5.04}$&0.369&13.203&10.930&10.295&9.881&\object{DENIS J115542.9-222458} $^5$\\ 
\object{SIPS1351-1758} &13 51 57.20 $-$17 58 49.1&18.18$\pm^{10.85}_{6.80}$&0.222&12.815&10.823&10.210&9.927&\object{LP 798-49}\\ 
\object{SIPS0931-1717} &09 31 22.41 $-$17 17 41.8&17.46$\pm^{10.43}_{6.53}$&0.359&13.140&11.073&10.467&10.069&\object{DENIS J093122.3-171742} $^5$\\ 
\object{SIPS0435-1607} &04 35 15.97 $-$16 07 02.0&14.25$\pm^{8.55}_{5.34}$&0.355&12.341&10.406&9.779&9.352&\object{LP 775-31}\\ 
\object{SIPS0440-0530} &04 40 23.32 $-$05 30 07.8&10.21$\pm^{6.10}_{3.82}$&0.356&13.171&10.658&9.986&9.545&\object{LP 655-48}\\ 
\object{SIPS1324-0504} &13 24 46.44 $-$05 04 17.7&17.18$\pm^{10.25}_{6.42}$&0.328&11.043&9.465&8.861&8.563&\object{G 14-52} $^6$\\ 
\object{SIPS1607-0442} &16 07 31.25 $-$04 42 06.3&15.06$\pm^{8.99}_{5.63}$&0.462&14.866&11.896&11.187&10.717&\object{2MASSW J1607312-044209} $^7$\\ 
\object{SIPS0109-0343} &01 09 51.04 $-$03 43 26.3&12.74$\pm^{7.60}_{4.76}$&0.376&14.768&11.694&10.931&10.428&\object{LP 647-13}\\ 
\object{SIPS1712-0323} &17 12 04.49 $-$03 23 28.9&15.13$\pm^{9.02}_{5.65}$&0.385&14.482&11.607&10.994&10.637&\\ 
\object{SIPS1614-0251} &16 14 25.20 $-$02 51 03.5&16.73$\pm^{9.98}_{6.25}$&0.350&13.528&11.303&10.683&10.280&\object{LP 624-54}\\ 
\object{SIPS0539-0059} &05 39 51.86 $-$00 59 05.2&11.29$\pm^{7.68}_{4.57}$&0.363&17.836&14.033&13.104&12.527&\object{SDSS J053951.9-005901} $^8$\\ 
\hline
\end{tabular}}
\medskip
\par
Citation key: $^1$ Subasavage et~al. (\cite{Subasavage05}), $^2$ Kendall et~al. (\cite{Kendall06}), $^3$ Webb et~al. (\cite{Webb99}), $^4$ Reyle et~al. (\cite{Reyle02}), $^5$ Phan~Bao et~al. (\cite{Phan03}), $^6$ Giclas et~al. (\cite{Giclas71}), $^7$ Gizis et~al. (2002), $^8$ Fan et~al. (\cite{Fan00}). For objects designated NLTT or LP see Luyten's New Luyten Two Tenths catalogue (Luyten \cite{Luyten79}), for objects marked CE see Ruiz et~al. (\cite{Ruiz01}).
\end{table}